RICH1 Phone Meeting Friday 23rd November 2007
Present

CERN: Fabio Metlica, Olav Ullaland, Ken Wyllie, Werner Witzeling, Tatsuya Nakada, Cristophe Frei

Imperial College:  Bill Cameron, Richard Plackett, Dave Websdale

Oxford: Rohan Senanayake, Neville Harnew

RAL: Derek Morrow
Bristol: Richard Head

Apologies : Nick Brook

Mirrors
The CMA sample mirrors have been taken out of the C4F10 tank after 17 months and they are showing no change to D0 and R within errors.

Mechanical and optical tests will be completed next week

There has been no change to the irradiated sample after a year.

The tests were done at ambient temperature.

In short there has been no degradation in the test samples.

Lower Shield
(Context) Measurements indicate that the lowers shield moved during the magnet tests.

Photographs have been taken of the mounts. No decision has been made yet on what corrective measures should be taken.  The timescale is to secure the lower shield by the end of the year.  

One possibility is to add fixings at the corners of the shield to provide more of a lever.

WW has a meeting to discuss the options on Monday (26/11/07)

Gas Lines
There is a discussion of the routing of the C4F10 gas lines.

Christophe has some proposals. The upper system mounted on the wall – no problem for the RICH systems.

The lower system on the C side will be below the lower floor at the level of the cable trays. It will be clear of the magnetic doors when they are removed.

The flexible pipe will start ~1m from the shielding. We will wait until the ridged pipe is installed before installing the flexible pipe.

HPD Mechanics
Upper Rail system
The new support beam is at Oxford. RS has seen it but not checked it.
Lower Rail System
The conceptual design will be ready next week. There will be a meeting between IC and Oxford to check it.  BC etc also need to speak to Tony Handford about the extraction frame. There are some problems with the ladder extraction frame concept.

The design of the rail system will be circulated by RS on Monday. (It was received).
Upper HPD Box
Still disappointing progress

Workshop has 11 drawings of which 4 are complete.  This batch will take 2 weeks to finish. Five pieces have been sent to outside companies and will take a week to delivery. Some small parts are modelled but still need detailed drawings.

There will be an additional delay due to anodizing.

We are expecting that it will still take 4 weeks until assembly

Trevor is now working on the Photon Funnel; he will complete the remaining HPD box drawings as the workshop requires them.
Upper Box Handling
There has been an offer from Bristol to help design the HPD box handling equipment.

The lifting beam at CERN is identified – RH looking at the interface.

The beam seems to be slightly too short for the box.

A design for a lifting frame was presented although given the angles involved it is probably going to be more complicated to manufacture than a new lifting beam.

The design of photon funnel installation equipment could be handled by Oxford.

NH requests a list of standalone items that they could help with.

Quartz Windows
DM will visit CERN on 12/12/07 to organise lab space and equipment.
We should try and glue the pieces of the quartz windows together before they are bonded into the frame. This will save time between the PF delivery and potting the quartz in.

IC will provide an update of PF design changes as soon as it is available.

Cabling
No change; waiting for the first HPD box.
DW reports that one of the LV cables seems a bit short so we will probably need to pull it through a little from the barracks.
We will need to put in some scaffolding for the cable tray installation.

When the scaffolding is in place it would be an opportunity to finish the Data fibre installation.

*Discussion of Fibre installation*
Christophe will prepare a proposal (which has been circulated).
Syracuse Mag-Cal cables are somewhat undefined.
Long distance cabling to the counting rooms will be closed soon when the wall is completed.

Ancillary Systems
Light leak detectors – one PMT holder ready.
Temp sensors becoming critical.
Time alignment system will use a spare data fibre breakout bundle.

Photon Funnel
There has been MDM/Mag Cal progress.

The cable management is still missing but other components are progressing.

There is a 3D model from Syracuse.

Trevor is working on it now. We currently have sufficient information from Syracuse to proceed. Trevor has completed a model of the actual space the PF has to go in to. The next step is to check this against survey points.

We should have the space to +/-3mm.

The PF will be redesigned to fit the 3D space available in reality and the Syracuse CAD file. Trevor hopes to know if it’s possible by the end of next week.

It will be made by a company on whom Imperial can place orders directly and we expect it to take ~5 weeks if we pre-order the material.

LAMS
No change since last time

HPD Columns
Ken will upgrade the SPECS, and might add firmware upgrade cables as they won’t interfere with anything and will make further upgrades much easier.

Minutes by Richard Plackett
